Author’s response to reviews

Title: Decreased levels of physical activity in adolescents with Down syndrome are related with low bone mineral density: a cross-sectional study

Authors:

Ángel Matute-Llorente (amatute@gmail.com)
Alejandro González-Agüero (alg28@aber.ac.uk)
Alba Gómez-Cabello (agomez@unizar.es)
Germán Vicente-Rodríguez (gervicen@unizar.es)
José Antonio Casajús (joseant@unizar.es)

Version: 4 Date: 24 June 2013

Author’s response to reviews: see over
Dear Editor,

Please, find attached the latest version of our manuscript entitled “Decreased levels of physical activity in adolescents with Down syndrome are related with low bone mineral density: a cross-sectional study” to be considered for publication in your prestigious journal.

This manuscript represents original unpublished material; is not under consideration for publication elsewhere, and further, it will not be submitted for publication elsewhere until a decision is made regarding its acceptability for publication in your Journal.

All authors have read and approved of this final version, are responsible for the reported research and have contributed significantly to the research of the present manuscript.

There are no conflicts of interest and financial disclosures for any author of this manuscript.

None of the authors have any financial interest.

Sincerely yours,

On behalf of all the co-authors

Ángel Matute-Llorente